

# Investigating the impact of CO2 on lowfrequency variability of the AMOC in HadCM3

Article

Published Version

Creative Commons: Attribution 4.0 (CC-BY)

Open access

Armstrong, E., Valdes, P., House, J. and Singarayer, J. (2017) Investigating the impact of CO2 on low-frequency variability of the AMOC in HadCM3. Journal of Climate, 30 (19). pp. 7863-7883. ISSN 1520-0442 doi: 10.1175/jcli-d-16-0767.1 Available at https://centaur.reading.ac.uk/73032/

It is advisable to refer to the publisher's version if you intend to cite from the work. See <u>Guidance on citing</u>.

To link to this article DOI: http://dx.doi.org/10.1175/jcli-d-16-0767.1

Publisher: American Meteorological Society

All outputs in CentAUR are protected by Intellectual Property Rights law, including copyright law. Copyright and IPR is retained by the creators or other copyright holders. Terms and conditions for use of this material are defined in the <u>End User Agreement</u>.

www.reading.ac.uk/centaur

CentAUR



# Central Archive at the University of Reading

Reading's research outputs online

# <sup>a</sup>Investigating the Impact of CO<sub>2</sub> on Low-Frequency Variability of the AMOC in HadCM3

EDWARD ARMSTRONG, PAUL VALDES, AND JO HOUSE

School of Geographical Sciences, University of Bristol, United Kingdom

#### JOY SINGARAYER

Centre for Past Climate Change, and Department of Meteorology, University of Reading, Reading, United Kingdom

(Manuscript received 24 October 2016, in final form 8 June 2017)

#### ABSTRACT

This study investigates the impact of CO<sub>2</sub> on the amplitude, frequency, and mechanisms of Atlantic meridional overturning circulation (AMOC) variability in millennial simulations of the HadCM3 coupled climate model. Multichannel singular spectrum analysis (MSSA) and empirical orthogonal functions (EOFs) are applied to the AMOC at four quasi-equilibrium  $CO_2$  forcings. The amount of variance explained by the first and second eigenmodes appears to be small (i.e., 11.19%); however, the results indicate that both AMOC strength and variability weaken at higher CO<sub>2</sub> concentrations. This accompanies an apparent shift from a predominant 100-125-yr cycle at 350 ppm to 160 yr at 1400 ppm. Changes in amplitude are shown to feed back onto the atmosphere. Variability may be linked to salinity-driven density changes in the Greenland-Iceland-Norwegian Seas, fueled by advection of anomalies predominantly from the Arctic and Caribbean regions. A positive density anomaly accompanies a decrease in stratification and an increase in convection and Ekman pumping, generating a strong phase of the AMOC (and vice versa). Arctic anomalies may be generated via an internal ocean mode that may be key in driving variability and are shown to weaken at higher CO<sub>2</sub>, possibly driving the overall reduction in amplitude. Tropical anomalies may play a secondary role in modulating variability and are thought to be more influential at higher CO<sub>2</sub>, possibly due to an increased residence time in the subtropical gyre and/or increased surface runoff driven by simulated dieback of the Amazon rain forest. These results indicate that  $CO_2$  may not only weaken AMOC strength but also alter the mechanisms that drive variability, both of which have implications for climate change on multicentury time scales.

#### 1. Introduction

The Atlantic meridional overturning circulation (AMOC) is an important system of ocean currents that transport significant quantities of heat across the North Atlantic. It has important implications for Northern Hemisphere climate, with the potential to influence surface air temperatures (SATs), sea surface temperatures (SSTs), and precipitation (e.g., Vellinga and Wu 2004; Knight 2005; Frankcombe and Dijkstra 2010; Delworth and Mann 2000).

DOI: 10.1175/JCLI-D-16-0767.1

© 2017 American Meteorological Society

Both observational (Clark et al. 2002) and modeling studies (Swingedouw et al. 2014) have shown the potential for the AMOC to fluctuate on strength on a range of time scales. In the absence of long observational datasets (Wunsch and Heimbach 2006), climate models are used in order to understand longer time scales of variability (e.g., von Storch et al. 2000). Simulations forced with current and preindustrial concentrations have yielded a range of spectral time scales from decadal to centennial depending on the model used (see Swingedouw et al. 2014). A number of mechanisms have been attributed to driving this variability. Models such as IPSL (Msadek and Frankignoul 2009), CCSM (Danabasoglu 2008; Danabasoglu et al. 2012), and ECHAM5 (Timmermann



This article is licensed under a Creative Commons Attribution 4.0 license (http://creativecommons. org/licenses/by/4.0/).

<sup>&</sup>lt;sup>o</sup> Denotes content that is immediately available upon publication as open access.

Corresponding author: Edward Armstrong, edward.armstrong@ bristol.ac.uk

et al. 1998; Zhu and Jungclaus 2008) indicate a two-way coupling between the atmosphere and ocean. Studies using the GFDL model (Delworth et al. 1993; Delworth and Greatbatch 2000; Delworth et al. 1997) and ECHAM5–MPI-OM (Jungclaus et al. 2005) concluded that fluctuations are oceanic in origin but forced by the atmosphere. External forcing events such as volcanic eruptions (Otterå et al. 2010; Zanchettin et al. 2012; Iwi et al. 2012; Stenchikov et al. 2009) and salinity anomalies induced by El Niño (Mignot and Frankignoul 2005; Schmittner et al. 2000) have also been suggested.

The potential for the AMOC to influence regional and global climate has prompted a range of modeling studies investigating how it may respond to anthropogenic climate change. These have commonly focused on the impact of  $CO_2$  on overall AMOC strength, with a number of studies showing that higher concentrations will weaken the AMOC, albeit to varying degrees (Gregory et al. 2005). This has primarily been linked to a reduction in surface density in sinking regions, via surface heat loss and/or a change in the freshwater flux (e.g., Gregory et al. 2005; Swingedouw et al. 2007; Thorpe et al. 2001; Dixon et al. 1999; Mikolajewicz and Voss 2000; Bakker et al. 2016b). In contrast, there has been less focus on the possible effects of  $CO_2$  on the amplitude and frequency of AMOC variability.

This study will investigate how  $CO_2$  concentration impacts low-frequency variability of the AMOC in the HadCM3 coupled climate model. Although there are clear limitations to using one model, it does allow thorough investigation of the potential mechanisms that would not be possible with a multimodel study. Despite a number of drawbacks (see section 2), HadCM3 has been shown to simulate the AMOC, ocean heat transport and aspects of the freshwater ocean– atmosphere cycle accurately compared to observations (Gordon et al. 2000; Pardaens et al. 2003). Furthermore the resolution of the model permits millennial-scale simulations in order to study low-frequency variability.

Previous studies with HadCM3 have highlighted the relationship between density anomalies in the Greenland–Iceland–Norwegian (GIN) Seas and AMOC variability. Variations in density have commonly been linked to salinity anomalies, which are advected into the region primarily from the Arctic and/or Caribbean regions. Vellinga and Wu (2004, hereafter VW04) identified a coupled ocean–atmosphere mechanism in the tropics, with periods of strong AMOC driving an increased equatorial SST gradient, a northward shift in the position of the ITCZ, and an increase in precipitation north of the equator. This generates a fresh anomaly that is advected north into the region of downwelling initiating a reversal in AMOC strength and vice versa. This observation follows earlier work by Manabe and Stouffer (1997), who showed via a hosing experiment that the AMOC was sensitive to subtropical freshening. Menary et al. (2012) showed that this mechanism was also present in the Kiel Climate model and highlighted proxy records that give a comparable periodicity of between 50 and 150 yr as evidence for the mechanism.

Hawkins and Sutton (2007, hereafter HS07) used 3D empirical orthogonal functions (EOFs) to highlight the importance of the Arctic, which releases salinity anomalies into the GIN Seas, increasing convection and possibly driving an increase in AMOC strength. The formation of these anomalies was hypothesized to be a result of an internal ocean mode controlled by salinity gradients between the Arctic and the GIN Seas, with southward advection in Arctic surface waters compensated by deep northward advection of an opposing anomaly from the GIN Seas.

Jackson and Vellinga (2013, hereafter JV13) concluded that both the mechanisms of VW04 and HS07 might be present, with the Arctic playing a more predominant role and the Caribbean region modulating variability when it is present. In contrast to HS07, they deduced that Arctic anomaly formation is a coupled atmosphere–ocean feedback, responding to stochastic sea level pressure (SLP) variations. This ageostrophically alters the geostrophic balance in the Beaufort Gyre, driving upwelling and advection of salinity into the GIN Seas. A perturbed physics ensemble that altered surface fluxes and winds showed that the background climate might also be important in determining both the frequency and amplitude of variability.

Although the amplitude, frequency, mechanisms, and impacts of AMOC variability have been well researched, there has been less focus on how these may be altered at higher CO<sub>2</sub> concentrations. A number of previous studies that have investigated time-dependent dynamic systems such as the AMOC have focused on the framework introduced in the Stommel box model (Stommel 1961), which indicates that a decline in the strength of the system, such as that induced by higher CO<sub>2</sub>, would increase instability, pushing the system closer to a point of collapse (i.e., a bifurcation point) and so increase variability (Tziperman 1997; Tziperman et al. 1994; Wiesenfeld and Mcnamara 1986; Ditlevsen and Johnsen 2010; Rahmstorf 1995; Monahan 2002). However, results from MacMartin et al. (2016) were not consistent with this hypothesis. They investigated the impact of  $CO_2$  on the AMOC at a range of latitudes in two simulations of GFDL-ESM2M. They showed that at  $4 \times CO_2$  a decrease in AMOC strength was accompanied by a decrease in midlatitude variability, with the region of greatest variability shifted to a limited area at higher latitudes. This was linked to a

change in ocean dynamics, with an increase in stratification possibly reflecting a weaker and shallower mean overturning at low latitudes.

In this study we aim to answer two key questions: 1) Does  $CO_2$  impact the amplitude/frequency of AMOC variability? 2) What mechanisms are responsible for these changes? Section 2 outlines the model and methods and section 3 gives an overview of the climatology and impact of  $CO_2$  on AMOC strength. Section 4 investigates spatial and time series variability, and section 5 discusses the climatic implications of variability. Section 6 examines the possible mechanisms before a discussion and summary is presented in section 7.

#### 2. Methods

#### a. Model

The Hadley Centre Climate Model, version 3 (HadCM3), is a coupled Earth system model consisting of 3D dynamical atmosphere and ocean components (Valdes et al. 2017). The atmospheric component has a horizontal resolution of  $3.75^{\circ} \times 2.5^{\circ}$  with 19 vertical levels. The ocean has a horizontal resolution of  $1.25^{\circ} \times$ 1.25° with 20 vertical levels that have a finer resolution toward the surface. The ocean uses a "rigid lid" approach so volume remains constant, with runoff from the land converted to salinity flux where river outflow points reach the ocean. The mixed layer model in the ocean is that of Kraus and Turner (1967) with vertical mixing of tracers using a simplified version of the Large et al. (1994) scheme, which is discussed in detail in Gordon et al. (2000). Horizontal mixing of tracers (e.g., salinity) via eddies uses the Gent and McWilliams (1990) isopycnal parameterization with isopycnal mixing using the Redi (1982) scheme implemented by the method of Griffies et al. (1998). HadCM3 does not use flux adjustment, which can impact deep-water formation and influence the AMOC (Marotzke and Stone 1995).

The drawbacks of the model include issues related to ocean resolution that affects some significant channels including the North Atlantic overflows (Roberts et al. 1996). The Canadian Archipelago, including the Hudson and Davis Straits, is closed and there is zero barotropic flow through the Bering Strait and so no net volume transport. This is in disagreement with observations that indicate that freshwater import into the Arctic via the Bering Strait is significant (Cattle and Cresswell 2000). To compensate for this, all freshwater export from the Arctic is via the Atlantic sector into the Nordic seas. This may have implications for the AMOC as the Bering Strait region has been shown to moderate AMOC strength (Hu and Meehl 2005). The Denmark Strait and Greenland– Scotland ridge have been deepened in order to produce a mean outflow that matches observations (Gordon et al. 2000). Other problems include a weakness in the wind stress in the North Atlantic storm track, which potentially impacts gyre strength (Gordon et al. 2000). Furthermore, peak flow of the AMOC in HadCM3 is at approximately 800 m (see Fig. 4), shallower than the 1000 m observed by RAPID-MOCHA (Smeed et al. 2015). The cause of this is likely to be surplus surface salinity in the North Atlantic, driven by an excess in net evaporation in the topics and subtropics and to a lesser extent insufficient subtropical runoff (Pardaens et al. 2003). Consequently the Atlantic is too stratified and stable, reducing the depth of maximum overturning. This has also been shown to be the case in other models, including GFDL CM2.1, NCAR CCSM3, and the MPI models (Roberts et al. 2013; Msadek et al. 2013).

The results presented here are from 2000-yr quasiequilibrium simulations at four CO<sub>2</sub> concentrations; 350, 700, 1050, and 1400 ppm. We will refer to these experiments as 1x, 2x, 3x, and 4x, respectively. All analysis is conducted on the final 1000 yr of each simulation, a time period that is suitable for examining low-frequency variability. It also permits a 1000-yr spinup in order for the ocean to reach a relative state of equilibrium as indicated by the volume averaged upper ocean temperatures (not shown). The full time series of the AMOC index (i.e., mean AMOC strength between 40° and 50°N at 800 m; see section 4b) for each experiment are shown in Fig. 1, showing that the AMOC has reached a relative state of equilibrium in each simulation after 1000 yr. To remove a small climate drift the AMOC streamfunction and climate variables have been detrended by subtracting a liner least squares fit prior to analysis.

### b. Statistics

#### 1) SINGULAR SPECTRUM ANALYSIS

To analyze variability of the AMOC we have employed singular spectrum analysis (SSA) and its multivariate counterpart (MSSA). This is a nonparametric technique used to decompose a time series into a range of periodicities and noise in order to extract a signal of interest and more clearly understand the dynamical nature of the system being studied. The MSSA technique can be used to analyze gridded datasets. The methodologies and applications of SSA and MSSA are outlined in Vautard et al. (1992), Elsner and Tsonis (1996), and, more recently, Golyandina and Zhigljavsky (2013), with a thorough mathematical overview for the application to climatic time series in Ghil et al. (2002). Numerous studies have applied SSA and MSSA to climatological data (e.g., Ghil et al. 2002; Plaut et al. 1995; Moron et al. 1998), including the AMOC (Alvarez-Garcia et al. 2008;



FIG. 1. Time series of the AMOC index for the full 2000-yr simulation at (top)–(bottom) 1x, 2x, 3x, and 4x. The AMOC index is defined as the mean annual strength of the AMOC between  $40^{\circ}$  and  $50^{\circ}$ N at 800 m. The final 1000 yr of each simulation are used in this study.

Huang et al. 2012). The SSA and MSSA in this paper have utilized the R package "Rssa," which is extensively outlined in Golyandina and Korobeynikov (2014) and Golyandina et al. (2015).

The SSA algorithm consists of two basic stages, decomposition and reconstruction. First a covariance matrix is generated from the time series with a specified window length representing the number of lags M. Consequently the covariance matrix is of dimensions  $M \times M$ . MSSA differs from SSA in the way this trajectory matrix is constructed (see Golyandina et al. 2015). The matrix undergoes singular value decomposition (SVD) to produce corresponding eigenvectors and eigenvalues. The projection of the original time series onto the EOFs yields the principal components (PCs), with the first PCs (PC1s) representing the largest amount of variance. These PCs can be "reconstructed" and put back on the scale of the original time series by projecting back onto the corresponding eigenvectors. Combination of all the individual reconstructed PCs would generate the original time series; however, partial reconstruction permits the identification of potential oscillatory components and the removal of noise.

An important consideration in SSA and MSSA is the window length M. Ghil et al. (2002) state that the size of *M* is a compromise: a large *M* allows more information to be extracted from the original time series; however, it will permit fewer repetitions, which consequently reduces statistical confidence. Other studies state that Mshould be approximately divisible by the time scale of a potential oscillation in the data (Golyandina 2010; Golyandina et al. 2001). In this study, we tested AMOC variability for a range of window lengths ranging from 100 to 300 yr. The results were similar for each value of *M* indicating stability in our results; the values used are outlined in sections 5a and 5b. All climate variables used in this study have been decomposed using the MSSA technique with a window length that corresponds to the appropriate CO<sub>2</sub> concentration.

#### 2) SIGNIFICANCE

To calculate the statistical significance of the anomalies and correlations we have used the moving block bootstrap technique (Wilks 1997) with 95% confidence limits calculated using the bootstrap percentile method (see Hall 1988). If the bootstrap confidence interval passes through zero the anomaly/correlation is also deemed insignificant. Power spectra have been generated using discrete Fourier transform to reveal periodicities and their relative strengths. We have tested the significance of these peaks against the 95% confidence level determined by a red noise spectrum of a first-order autoregressive (AR1) process.

## 3. Impact of CO<sub>2</sub>

## a. Climatology

The impact of  $CO_2$  on a range of climate variables is shown in Fig. 2. SATs and SSTs are amplified across the region, the former predominantly at high latitudes and over land and the latter to the north of Scandinavia and east of the United States. There is a complex pattern of precipitation change, with a northward shift and/or weakening of the ITCZ driving a decrease across much of the tropics and an increase at high latitudes. This pattern of temperature and precipitation changes results in an increase in salinity across the tropics and a decrease at the poles. There is a dramatic decrease in sea ice, with almost complete disappearance at  $4 \times CO_2$ across much of the Arctic. Many of the climate impacts simulated by HadCM3 are consistent with those from other climate models, as shown in the CMIP5 experiments (e.g., Knutti and Sedlacek 2013) and the IPCC Fifth Assessment Report (AR5; Flato et al. 2013).

The mixed layer climatology gives a guide as to the key location of deep-water formation. The GIN Seas

ARMSTRONG ET AL.



FIG. 2. Mean climatologies for (a) 1x and anomalies at (b) 2x, (c) 3x, and (d) 4x for (left)–(right) precipitation, SAT, salinity, the mixed layer depth, sea ice, and SST. Only anomalies that are considered 95% confident are shown (see section 2).

region shows the greatest depth of overturning extending to mean depths of 247 m off the coast of Scandinavia. In contrast to other models (such as CCSM), HadCM3 exhibits shallow overturning depths in the Labrador Sea. There is a reduction in the depth of the mixed layer in the GIN Seas of up to 87 m at  $4 \times CO_2$ .

Mean depth profiles for salinity, temperature, and density for the GIN Seas (see Table 1; Fig. 11a) are shown in Fig. 3. For reference these are shown with the NODC Levitus observational dataset spanning 1900–92. Within the model, cooler, fresher, and less dense surface waters overlie warmer, saltier, and denser subsurface waters. Ocean velocity maps (not shown) show that this stratification is likely to be a result of a warmer northeastward flow of water via the North Atlantic Current (NAC), which sits below colder fresher water transported southward by the East Greenland Current (EGC).  $CO_2$  alters these

profiles by increasing temperature and density throughout the water column. Salinity decreases at the surface and increases in the subsurface, the former possibly reflecting the modeled increase in precipitation in the high northern latitudes and the latter reflecting saltier Atlantic inflow at higher  $CO_2$ . Furthermore the water column is more stratified, indicated by greater contrast between surface and subsurface temperature and salinity.

TABLE 1. Latitudes and longitudes for the regions used in
this study.

Region	Lat	Lon
GIN Seas	60.625°-79.375°N	15°W–10°E
Arctic	84.375°–89.375°N	178.75°E–180°
EGC	65.625°-80.625°N	25°-15°W
Caribbean	10.625°-30.625°N	90°–50°W



FIG. 3. Mean depth profiles for potential temperature (solid; °C), salinity (dashed), and density (kg m<sup>-2</sup>; dotted) in the GIN Seas (see Table 1) for the different simulations. The NODC Levitus observational dataset is included for reference, showing mean values spanning 1900–92 (data provided by the NOAA/OAR/ESRL PSD, Boulder, Colorado, from their website at http://www.esrl.noaa.gov/psd/). The GIN Seas region in the Levitus data is defined as 60.625°-79.375°N, 13.75°W-11.25°E.

#### b. AMOC strength

The mean Atlantic AMOC streamfunction for 1x and the anomalies due to increasing CO<sub>2</sub> are shown in Fig. 4. At 1x, the upper North Atlantic Deep Water (NADW) cell reaches a peak of 16.6 Sv ( $1 \text{ Sv} \equiv 10^6 \text{ m}^3 \text{ s}^{-1}$ ) at 35°N at a depth of 800 m. The mean strength compares well to the 16.9 Sv observed by the RAPID–MOCHA (between April 2004 and October 2015 at 26.5°N) but is too shallow compared to observations (Smeed et al. 2015). There is a reduction in AMOC strength across this cell of up to 5.0 Sv (30.2%) at 2x, 4.95 Sv (29.8%) at 3x, and 5.6 Sv (34%) at 4x, in addition to a potential shallowing in the NADW cell. There is also a decrease in the strength of the northward moving Antarctic Bottom Water (AABW) cell and surface waters (0°–30°S and 30°N) of up to 3.8 Sv.

The decline in AMOC strength has been commonly linked to changes in surface heat loss and/or the freshwater flux that reduces density in downwelling regions. (e.g., Gregory et al. 2005; Swingedouw et al. 2007; Thorpe et al. 2001; Dixon et al. 1999; Mikolajewicz and Voss 2000). Figure 3 shows the reduction in density in



FIG. 4. (a) Mean AMOC streamfunction (Sv) at 1x for the final 1000 yr of the simulation and the anomalies at (b) 2x, (c) 3x, and (d) 4x. Only anomalies that are considered 95% confident are shown (see section 2).

the GIN Seas at higher  $CO_2$ . Thorpe et al. (2001) and Thorpe (2005) have previously investigated  $CO_2$ -induced weakening of the AMOC in HadCM3. They concluded that both  $CO_2$ -induced warming and a decrease in salinity at high latitudes drive this change, with the former playing a more prominent role accounting for approximately 60% of weakening.

It is also important to highlight the increase in stratification in the GIN Seas at higher  $CO_2$  (Fig. 3), which would consequently decrease convection and may weaken overturning. This is highlighted by the Brunt–Väisälä (or buoyancy) frequency for the top





FIG. 5. Brunt–Väisälä (or buoyancy) frequency in the GIN Seas for the top 666 m at (a) 1x and anomalies due to increasing  $CO_2$  at (b) 2x, (c) 3x, and (d) 4x. High values indicate a more stable water column and thus reduced convection. Only anomalies that are considered 95% confident are shown (see section 2).

660 m at 1x and the anomalies due to increasing  $CO_2$ (Fig. 5), which give an indication of the stability of the water column and possible levels of convection. Large values indicate a greater density gradient, more stable conditions, and a reduced rate of convection. At 1x the lowest values and so a region of weaker stability is found around the GIN Seas region with values increasing at higher  $CO_2$  specifically in the center of the region, indicating greater stability and thus reduced convection. This may act to reduce crossisopycnal flow and drive the apparent reduction in AMOC strength.

# 4. The variability of the AMOC

#### a. Spatial variability

To understand the spatiotemporal patterns of AMOC variance, we have applied MSSA (see section 2b) to the AMOC streamfunction for the region 30°S-90°N. This procedure was originally outlined in Plaut and Vautard (1994) and subsequently used in Huang et al. (2012). The AMOC streamfunction was first decomposed using MSSA with a range of M values from 100 to 300 yr (see section 2), with similar results produced. At 1x, 2x, and 3x values, the first and second eigenmodes bear the resemblance of a sine and cosine pair with a similar period that is out of phase (i.e., a phase quadrature relationship) and represent an oscillation on the order of approximately 110-125 yr. We chose an M value of 115 yr for 1x and 125 yr for 2x and 3x. At 4x the oscillation for the first and second is on the order of 150-160 yr and we chose an M value of 160 yr. These values for M were also chosen based on the initial spectral analysis of the AMOC index (as shown in Fig. 7 and discussed in section 4b).

The leading two eigenmodes are used to reconstruct the AMOC streamfunction. Together they account for 11.19% (6.86% and 6.69%), 7.69% (5.36% and 4.99%), 5.51% (3.84% and 3.8%), and 3.88% (2.8% and 2.3%) of the total variance for 1x, 2x, 3x, and 4x respectively. Although these values appear to be small, at 1x they are similar to those presented in previous studies that have derived oscillating MSSA modes for the AMOC (Huang et al. 2012) and for SST variability of the world oceans (Moron et al. 1998). The similarity of the first and second modes and 2D graphs of the eigenvectors (not shown) indicate a degenerate pair at all concentrations albeit more weakly at higher CO<sub>2</sub> concentrations. There is also a relatively coherent degenerate pair oscillation for a combination of eigenmodes between 5 and 9 depending on the concentration of CO<sub>2</sub>. This is at a higher frequency and may indicate a weaker oscillation of between 15 and 20 yr, possibly the frequency identified in the study of Dong and Sutton (2005). These eigenmodes will be investigated in a future study. We have applied this MSSA method and these values of *M* to all climate variables used in this study.

To identify the key areas of variance, EOF analysis is applied to the MSSA AMOC streamfunction. The values of the EOFs and corresponding PCs are arbitrary, so we have regressed the MSSA time series onto the normalized PCs in order to put them on the same scale and allow comparison. The first and second EOFs and their corresponding scaled PCs are shown in Fig. 6. At 4x the EOFs are based on the final 800 yr of the model run as the magnitude was anomalously large in the first 200 years. The PCs show approximately nine oscillations at



FIG. 6. (a) (left) First and (right) second EOFs of the reconstructed two leading eigenmodes (amount of variance outlined in text) of the MSSA decomposed AMOC streamfunction, and the amount of variance they explain. (b) The corresponding PC1 and PC2 with the MOI (mean strength of the AMOC between 40° and 50°N at 800 m) and the SSA decomposed MOI (SSA MOI) reconstructed from the first and second eigenmodes as outlined in the text. The correlation between PC1 and the SSA MOI is given at the bottom of each panel; (top)– (bottom) 1x, 2x, 3x, and 4x, respectively.

1x, 2x, and 3x, equivalent to 111-yr cycles, and possibly five oscillations at 4x, equivalent to 160 yr.

At 1x, the first EOF explains 87.2% of the variance of the MSSA data, which decreases to 30.8% at 4x. The first EOF maps for all CO<sub>2</sub> concentrations are characterized by a singular deep overturning cell southward of 60°N that corresponds to basinwide variability of the AMOC. This peaks between 40° and 50°N at a depth of 800-1000 m for all CO<sub>2</sub> concentrations. The northward boundary of the cell is limited by the location of the sill on the southern border of the GIN Seas. North of this there appears to be two cells of different sign at 1x that may correspond to a north-south shift in variability within the GIN Seas and Arctic. The second EOF explains 3.2% of the variance of the AMOC at 1x increasing to 19.5% at 4x. This EOF is more complicated, showing positive and negative anomalies that may reflect variability in both the NADW and AABW cells.

At higher concentrations of  $CO_2$  the depth, southward extent, and strength of variance in the main cell decreases, accompanied by a reduction north of 60°N. The region of greatest variance remains at approximately 40°–50°N and 800-m depth for all simulations, in contrast to MacMartin et al. (2016), who identified a northward shift in variance at  $4 \times CO_2$ . The PC1 and PC2 time series indicate a decline in the clarity of oscillations and a possible shift toward lower frequency. At 1x and 2x AMOC amplitude and frequency is relatively coherent; however, at 3x there is a decline in amplitude between 375 and 800 yr and at 4x both frequency and amplitude appear relatively intermittent. This would indicate that higher concentrations of  $CO_2$  influence not only AMOC strength and depth of overturning, but also the amplitude, frequency, and coherence of variability.

#### b. Time series variability

To measure temporal variability, it is common to define an AMOC index [meridional overturning index (MOI)] at a latitude and depth where variability is greatest. Using the first EOFs as a guide, MOIs are calculated as the mean strength of the AMOC between 40° and 50°N at a depth of 800 m for all CO<sub>2</sub> concentrations. The MOI was decomposed using SSA (see section 2), using the same values for *M* (section 4a). As before, the procedure was tested with a range of window lengths and the results remained stable. Initial decomposition shows that the first



FIG. 7. Power spectra for (left) the original MOI and (right) the SSA decomposed MOI reconstructed from the eigenmodes outlined in the text for the four  $CO_2$  concentrations. The dashed lines represent the 95% confidence interval calculated from the original corresponding MOI.

eigentriple accounts for greater than 99% of the variance and corresponds to the overall trend of the time series. This trend does not contain oscillatory components and represents variability on longer time scales than those of interest in this study. To isolate higher-frequency variability, the overall trend is reconstructed and extracted and the resultant time series decomposed for a second time. The EOFs, eigenvalues, and 2D graphs of the eigenvectors (not shown) can be used to identify eigenvectors that may constitute an approximate sine wave with similar frequencies and a phase shift (see Golyandina and Korobeynikov 2014). The first and second eigenvectors are extracted and reconstructed for all  $CO_2$  concentrations and are shown with the original MOIs and the first and second PCs in Fig. 6. Together they account for 39.53% (21.18% and 18.35%), 31.02% (16.81% and 14.21%), 26.09% (14.33% and 11.76%), and 17.56% (9.15% and 8.41%) of the total variance for 1x, 2x, 3x, and 4x respectively. The correlation values between the MOI and PC1 are strong for each simulation, indicating that the MOI successfully represents variability within the overall AMOC streamfunction.

Power spectra for the original and the decomposed MOI time series are shown in Fig. 7. Spectral peaks are consistent for both MOIs, although variability at higher frequencies has been removed for the SSA MOI spectrum. The peaks are significant at 1x, 2x, and 3x, but at 4x the peak is only just discernible against the 95% confidence interval of red noise. At 1x, the key time scale of variability is on the order of 100–125 yr, in agreement with the studies of VW04, JV13, and Menary et al. (2012). With increasing CO<sub>2</sub> there is a decrease in the power of this variability and an apparent shift to lower frequency at 4x. Spectral peaks are on the order of 112, 125, and 165 yr for 2x, 3x, and 4x respectively. This reiterates what is shown by the EOF analysis; higher

concentrations of  $CO_2$  act to weaken and possibly decrease the frequency of AMOC variability.

These results show the potential for  $CO_2$  to influence the variability of the AMOC, a similar conclusion to that found by MacMartin et al. (2016) with the GFDL model. There are, however, disparities in the results between the two models, namely that MacMartin et al. (2016) focused on a decadal 15–20-yr oscillation and showed that the amplitude of variability remained consistent yet shifted northward from 25°–50° to 60°N at 4x. We do not see this northward shift and see a decline in the amplitude of variability and a possible change in frequency. However, both studies challenge the theory of the Stommel (1961) box model, which stipulates that a weaker and thus less stable AMOC should exhibit greater variability.

#### 5. Feedback of variability on climate

To understand how the AMOC feeds back on climate, Fig. 8 shows composite anomaly maps for a number of variables during periods of maximum AMOC (AMax) compared to their mean state. Periods of AMax are defined as the years where PC1 (Fig. 6) for each  $CO_2$  concentration is more than one standard deviation above the mean. A 10-yr running mean has been applied to the climate variables followed by decomposition and reconstruction via MSSA. Composite maps showing AMOC minimum (AMin) are not shown but broadly show the opposite pattern.

AMax coincides with a broad increase in SSTs north of the equator compared to the mean, peaking at approximately  $0.8^{\circ}$ C at 1x and  $0.3^{\circ}$ C at 4x. Anomalously warm temperatures are focused within and to the north of the GIN Seas, ranging from  $0.6^{\circ}$  to  $0.8^{\circ}$ C at 1x. In the Arctic and to the south of the equator, AMax coincides with a decrease in SSTs on the order of  $0.1^{\circ}$ – $0.4^{\circ}$ C,



FIG. 8. (left)–(right) Composite climate anomaly maps during periods of AMax averaged over periods where the PC1 of the AMOC streamfunction is one standard deviation above the mean for (top)–(bottom) 1x, 2x, 3x, and 4x.

although at 1x and 2x there are positive SATs anomalies in the Arctic. Beyond this, anomalies in SATs and SSTs broadly correlate, with positive SATs spread across much of the central North Atlantic with cooling south of the equator at 1x and to a lesser extent at 4x. At 1x North Atlantic warming is on the order of  $0.3^{\circ}$ – $0.5^{\circ}$ C, peaking in the northern GIN Seas region at 1.7°C. Despite the impact of AMax on SAT and SST, there is a negligible impact on sea ice concentrations (not shown). Periods of AMax are associated with a significant northward shift in the position of the ITCZ over the Atlantic at 1x, increasing (decreasing) precipitation north (south) of the equator. This shift was first observed in HadCM3 by VW04, who linked it to the generation of salinity anomalies in the tropics (see section 6b). There is also a weak trend for an increase in precipitation across much of the North Atlantic extending into the GIN Seas. The northward shift in the position of the



FIG. 9. Depth profiles for the GIN Seas showing anomalies in (left) potential temperature, (center) salinity, and (right) density during periods of AMax (solid) and AMin (dashed) for increasing concentrations of CO<sub>2</sub>. A 10-yr running mean and MSSA has been applied to the climate variables.

ITCZ is significantly weakened at higher  $CO_2$  concentrations, specifically at 2x.

The anomalously warm SSTs and SATs at AMax are likely to drive the dipole pattern observed in SLP anomalies that display a strong area of low pressure north of the equator and high pressure to the south. The strongest low pressure anomaly is on the order of 42 Pa located within and to the south of the GIN Seas. This low pressure anomaly is also present at 2x but it diminishes in strength and expands into the high northern latitudes at 3x and 4x.

# 6. Impact of CO<sub>2</sub> on mechanisms driving variability

#### a. Convection in the GIN Seas

The degree of stratification and convection in the GIN Seas region is likely to play a key role in driving

AMOC variability. As shown in Fig. 5, there is a reduction in the Brunt-Väisälä frequency in the GIN Seas with increasing CO<sub>2</sub> indicating weaker convection and so a reduced overall AMOC. We can also investigate the degree of convection between periods of AMax and AMin. Figure 9 shows the average depth profiles in the GIN Seas for temperature, salinity, and density anomalies during AMax and AMin, and Fig. 10 shows the wind stress curl regressed onto the MOI and salinity. At AMax, surface temperature anomalies are warmer, below which to approximately 500 m there is a small increase at 1x and a cooling at 2x, 3x, and 4x, before an increase between 500 and 2000 m. Salinity and density show a large positive anomaly to approximately 300 m and remain positive to approximately 2000 m. When comparing the first 1000 m of these anomaly profiles to the mean state of



FIG. 10. The regression coefficients for the regression of wind stress curl (N m<sup>-3</sup>) with the (top) MOI and (bottom) salinity in the GIN Seas, for (a) 1x, (b) 2x, (c) 3x, and (d) 4x. Only anomalies that are considered 95% confident are shown (see section 2).



FIG. 11. Correlation (no lag) of MSSA decomposed salinity anomalies with the MOI, for (a) 1x, (b) 2x, (c) 3x, and (d) 4x. In (a) the boxes outline the GIN Seas, Caribbean, EGC, and Arctic regions that are used in the lagged correlation analysis. Only anomalies that are considered 95% confident are shown (see section 2).

the GIN Seas in Fig. 3, the pattern of anomalies during AMax is broadly opposite to the mean water profiles, indicating a less stratified water column at AMax. The opposite effect is seen at AMin.

This reduction in stratification acts to enhance convection at AMax as shown by an increase in wind stress curl in the GIN Seas (Fig. 10), particularly at 1x. The increase in temperature and salinity and consequent negative SLP anomaly (Fig. 8) acts to drive anticyclonic winds, which, in addition to the Coriolis force, drives a divergence in the Ekman layer and upward Ekman pumping, reducing stratification and subsequently increasing convection. Rather than the initial driving force, this process may be a positive feedback cycle that responds to an initial increase in the AMOC strength and acts to further enhance convection and strengthen the AMOC.

At higher concentrations of  $CO_2$  there is smaller contrast in temperature, density, and salinity between AMax and AMin, particularly in the top 1000 m. As a result the water column is more stratified during AMax, which, as shown in Fig. 5, weakens convection at higher  $CO_2$ . There is also a possible shift in convection to a region just north of Iceland at 2x and 3x, an impact also shown in Fig. 5. This lessened convection may reflect a weaker overall AMOC at higher  $CO_2$ , which in turn reduces the impact on SSTs and SATs at AMax and consequently the increase in Ekman pumping, which is required to drive enhanced variability.

#### b. Impact of regional salinity anomalies

Although Ekman pumping may enhance convection in the GIN Seas that subsequently drives a period of AMax, temperature- and salinity-driven density changes are likely to initiate the shift between AMax and AMin. As highlighted by Thorpe et al. (2001), the mean strength of the AMOC is primarily influenced by temperature-driven density changes. However, previous studies have commonly focused on salinity anomalies as the key driver of variability (Jungclaus et al. 2005; Delworth et al. 1997) including within HadCM3 (Hawkins and Sutton 2008; JV13; VW04; Pardaens et al. 2003; Menary et al. 2012).

Figure 11 shows the no-lag correlation of MSSA salinity averaged over the top 666 m with the MOI. At 1x there is strong correlation in the North Atlantic and the GIN Seas and a strong negative correlation in the Arctic and the southwestern North Atlantic basin. At increasing CO<sub>2</sub>, this general pattern remains, with strong negative correlation in the tropics and Arctic, yet there is a weakening in the positive correlation in and around the GIN Seas. This is also highlighted in the lower panels in Fig. 10, which shows a positive relationship between Ekman pumping and salinity in the GIN Seas at 1x; this weakens at higher  $CO_2$  with an apparent southward shift to a region just north of Iceland. This section will investigate the changing roles of salinity anomalies at higher  $CO_2$ .

As discussed, the studies of VW04, HS07, and JV13 concluded that salinity anomalies are created and advected primarily from the Arctic and Caribbean regions via the EGC and NAC respectively and are the primary method by which salinity anomalies are created in the GIN Seas. JV13 showed via a salinity budget analysis that regional processes such as sea ice fluctuations and surface processes play only a small role. A number of previous studies have highlighted the potential for Southern Ocean processes to impact AMOC variability. Bakker et al. (2016a) showed that fluctuations in the discharge from the Antarctic ice sheets in response to subsurface ocean temperatures influenced centennial-scale climate variability. Studies have also linked AMOC variability to circumpolar wind stress (Toggweiler and Samuels 1993) and to Weddell Sea processes driven by sea ice fluctuations and deep ocean convection (Martin et al. 2013; Park and Latif 2008; Latif et al. 2013). We performed a lagged correlation analysis of Southern Ocean salinity with the MOI (not shown) but did not find any significant correlations; therefore we will focus predominantly on the Caribbean, Arctic, and GIN Seas region in this study.

Figure 12 shows the lagged correlations of salinity with the MOI for the GIN Seas, EGC, Caribbean, and Arctic (Table 1 and Fig. 11). Positive anomalies in the GIN Seas and EGC occur at approximately zero lag (i.e., at AMax), and salinity in the Caribbean region peaks at approximately 30–53 yr prior to AMax and in the Arctic between 37 and 50 yr (Table 2). Regional correlation shows consistent oscillatory patterns, with time periods between peaks of 110 and 130 yr, a comparable time scale of variability to the MOI (Fig. 7). The strength of correlations weakens at higher CO<sub>2</sub> specifically for the Arctic; however, the Caribbean remains relatively consistent throughout.

To determine the time scale of salinity fluctuations, an EOF analysis was applied to the MSSA decomposed salinity in the different regions and the leading PCs extracted and regressed onto salinity. Figure 13 shows the MOI time series and the first PCs for the Arctic and Caribbean regions. Here the PCs have been lagged with the year where salinity in the region reaches a peak prior to AMax as shown in Fig. 12 and Table 2. The corresponding spectral analysis for each region is shown in Fig. 14. Salinity fluctuations within the Arctic have a greater magnitude than in the Caribbean and show



FIG. 12. Lagged correlations of MSSA decomposed salinity anomalies with the MOI for regions identified in Table 1 and Fig. 11a, for (top)–(bottom) 1x, 2x, 3x, and 4x. Thick dashed lines highlight 95% confidence for all time series (see section 2). Salinity anomalies lead the MOI for negative lags.

strong correlation with the MOI at 1x and 2x that declines at higher  $CO_2$  particularly at 4x. At 3x, the decline in the extent of variability beyond 400 yr is accompanied by a reduction in variance within the Arctic. The power spectra for Arctic salinity show strong similarities with both the GIN Seas and the overall MOI spectra (Fig. 7), indicating that this region has a strong relationship with the overall time scale of AMOC variability. In contrast, the strength of variability within the Caribbean region remains relatively consistent and shows a small increase at 4x. The power spectrum shows a significant peak at

TABLE 2. The lag values (yr) for the generation of positive Arctic and Caribbean salinity anomalies prior to AMax. They have been identified from the lead–lag analysis shown in Fig. 12. They have been applied to the PCs for Arctic and Caribbean anomalies in Fig. 13.

Simulation	Arctic lag (yr)	Caribbean lag (yr)
1x	37	43
2x	45	30
3x	48	38
4x	50	53



FIG. 13. The MOI (black) and PC1 for salinity anomalies from the Arctic (blue) and Caribbean (red), for (top)–(bottom) 1x, 2x, 3x, and 4x. The PCs have been lagged with values identified in the lead–lag analysis in Fig. 12 and shown in Table 2.

approximately 160 yr at 4x, consistent with the MOI spectra. Variability at lower CO<sub>2</sub> concentrations shows peaks that are compatible with the MOI but are small in comparison to the Arctic.

The PC and spectral patterns may indicate that the Arctic is key in setting the amplitude and time scale of variability while the Caribbean plays a secondary role, an idea similarly put forward by JV13. At higher concentrations of  $CO_2$ , weaker variability in Arctic salinity may drive a reduction in the overall amplitude of AMOC fluctuations. In contrast, variability of salinity in the Caribbean remains consistent and may increase, indicating that this region is comparatively more important for driving variability at higher  $CO_2$ .

#### 1) ARCTIC ANOMALIES

To understand more clearly the development of salinity anomalies in the Arctic, Fig. 15 shows the lagged correlation of Arctic salinity with the MOI prior to and following AMax. At -60 yr during periods of AMin, an isolated positive anomaly develops in the Beaufort Gyre in the center of the basin. This anomaly strengthens to a peak at lag -40 yr (as shown in Fig. 12) before being advected out of the gyre as shown in lag -20 yr. At this time, a negative anomaly develops north of Siberia that intensifies and spreads across the Beaufort Gyre at lag zero and +20 yr during periods of AMax. This negative anomaly is advected out of the basin at lag +40 yr and may contribute in pushing the AMOC into a negative phase. At this time, a positive anomaly develops off the Siberian coast that intensifies at lag +60 yr and restarts the cycle. JV13 showed a similar pattern of Arctic salinity anomalies across the Beaufort Gyre in HadCM3. This cyclical generation of anomalies is present but weakens at higher concentrations of CO<sub>2</sub>, concurrent with that shown by the EOF, spectral, and lead–lag analysis.

There remains uncertainty as to how salinity anomalies are generated in the Arctic and consequently how this mechanism may weaken at higher CO<sub>2</sub>. A salinity budget analysis may help to identify the role of different inputs and how they are influenced at higher CO<sub>2</sub>; however, we do not have the correct diagnostics to carry out this type of analysis. JV13 performed such an analysis on a preindustrial simulation of HadCM3, concluding that both sea ice processes and surface flow play only a minor role in the generation of Arctic anomalies. We cannot state that this is the case at higher concentrations of CO<sub>2</sub>; however, we do see only trivial fluctuations in sea ice cover between AMax and AMin and almost zero ice cover at  $4 \times CO_2$ , which may indicate that sea ice plays only a small role.

JV13 linked the formation of these Arctic anomalies to stochastic variations in SLP that alter the geostrophic balance in the Beaufort gyre. A period of high (low) sea level pressure drives an increase (decrease) in anticyclonic wind stress, strengthening (weakening) the gyre and resulting in downwelling (upwelling) and freshening (salinification) in the center of the basin and upwelling (downwelling) and salinification (freshening) at the coasts. We correlated MSSA SLP with the MOI (not shown); however, correlations were weak with no consistent pattern in the results to agree with this hypothesis, specifically at 2x and 3x where the pattern of salinity anomalies remain strong.

The study of HS07 stipulated that fluctuations in the AMOC might act as a lagged positive feedback on generating salinity anomalies in the Arctic. They highlighted an opposing pattern of salinity anomalies seen at depth in the Arctic basin that may be generated by deep advection of an opposing water mass from the GIN Seas. This migrates to shallower depth following advection of the surface anomaly. A lead-lag correlation analysis of the MOI with Arctic salinity at a range of depths is shown in Fig. 16. It indicates a relatively clear out-ofphase relationship between anomalies at depth relative to those at the surface. At  $1 \times CO_2$ , a positive (negative) anomaly at 5 m (300 m) follow approximately 40 yr (30 yr) later from those at depth. This oscillatory nature is coherent at 1x, 2x, and 3x but weakens at 4x. This weakening is consistent with the idea that this is a positive feedback and a direct consequence of weaker overall variability; that is, weaker variability is due to weaker oscillations in GIN Seas salinity, which in turn



FIG. 14. Power spectra of the PC1 of the MSSA decomposed salinity anomalies averaged over the top 666 m, in different ocean regions. These regions are identified in Table 1 and Fig. 11a. Note the change in scale for the Arctic. The dashed lines represent the 95% confidence interval where all spectra are considered significant.

drives weaker northward advection of anomalies into the Arctic and thus inhibits the formation of surface anomalies that are then advected south and required to maintain variability. It remains uncertain, however, what the overall driver is that weakens this mechanism. It may be a response to an overall reduction in strength of the AMOC; however, further experiments are required in order to test this hypothesis.

#### 2) CARIBBEAN ANOMALIES

The study of VW04 previously linked the generation of negative (positive) anomalies in the Caribbean to the northward (southward) shift of the ITCZ and an increase (decrease) in precipitation north of the equator during AMax (AMin). The resulting anomaly is advected north and pushes the AMOC into an opposing phase. This northward shift in the ITCZ is apparent in the precipitation anomalies in Fig. 8, but it is less pronounced at higher concentrations of CO<sub>2</sub>. Despite this, evidence in Figs. 11–14 indicates that variability in Caribbean salinity remains consistent and may even increase in amplitude at 4x.

A possible driver of this may be linked to the residence time of salinity anomalies in the Caribbean region. VW04 concluded from a tracer experiment that salinity anomalies advected from the Caribbean had a long residence time in the subtropical gyre, with subduction to subsurface waters prior to northward advection into the GIN Seas. Higher concentrations of CO2 may influence this process in two ways; first the weaker AMOC may slow the rate of northward advection. Second, increased salinity in the subtropical gyre at higher CO<sub>2</sub> is accompanied by an increase in stratification, which may reduce the rate of vertical mixing, a similar impact to that observed in the perturbed physics ensemble of JV13. Consequently the residence time in the subtropical gyre is increased and, because of relatively high evaporation, the salinity anomalies are intensified. This may not only produce the comparatively large anomalies at higher CO<sub>2</sub>, but also the increased advection time may result in the apparent shift toward lower frequencies (Fig. 14).

In addition to this, Caribbean salinity anomalies may also be influenced by changes in surface runoff rate at higher concentrations of  $CO_2$ . Figure 17 shows the surface runoff anomaly during periods of AMax for the different simulations. There is a greater impact at higher concentrations of  $CO_2$ , where there is a positive anomaly toward the north of the continent and a negative anomaly in the central regions. This general pattern at 3x and 4x reflects the change in precipitation rates over



FIG. 15. Lagged correlations of MSSA decomposed Arctic salinity averaged over the top 666 m with the MOI. Years before a change in the MOI are shown above, for (top)–(bottom) 1x, 2x, 3x, and 4x. Only correlations that are considered 95% confident are shown (see section 2).

this region at AMax as shown in Fig. 8. However, at these higher  $CO_2$  concentrations there is an increase in the proportion of this precipitation that is converted to runoff simulated by the model (Fig. 17b). This may reflect shifting vegetation dynamics across the Amazon region (Fig. 17a), with a change from predominantly broadleaf and C<sub>4</sub> grasses at  $1 \times CO_2$  to bare soil at higher CO<sub>2</sub> concentrations. This dieback of the Amazon rain forest in HadCM3 is a feedback that has been investigated in a number of previous studies (Betts et al. 2004; Cox et al. 2000; Boulton et al. 2013) and reflects a "tipping point" at which drying out and increasing temperatures overcome CO<sub>2</sub> fertilization and inhibit vegetation growth. Removal of vegetation alters the partitioning of water, reducing evapotranspiration and increasing infiltration driving a positive surface runoff anomaly. The river routing scheme (Gordon et al. 2000) channels this surface runoff instantaneously into the ocean via river outflow points. The key river catchment in this region is the Amazon River, with an outflow point located in northeastern Brazil at approximately the region of strongest negative correlation in Fig. 11. The increase in runoff rate at this point may help to drive the fluctuations in Caribbean salinity that influence AMOC variability at higher CO<sub>2</sub>, despite the reduced impact on the ITCZ.

This potential mechanism indicates that in addition to tropical atmosphere-ocean feedbacks, AMOC variability may also be influenced by terrestrial-ocean feedbacks brought about by atmospheric changes (i.e., precipitation). This mechanism is strengthened at higher CO<sub>2</sub> due to Amazon dieback, which may act to counter the weaker role of the Arctic in driving AMOC variability. The potential for Amazon runoff to impact salinity and consequently the AMOC was also identified by Latif et al. (2000), albeit caused by changes in precipitation rather than vegetation. We are aware that Amazon dieback is an artifact of the model that reflects regional trends in climate change, specifically precipitation, that vary between GCMs (Schaller et al. 2011). HadCM3 is therefore an extreme case, with other GCMs more consistently simulating a drier seasonal environment that remains forested (Malhi et al. 2009). However, the potential feedbacks of the land surface may have implications for predicting variability in climate models under increasing CO<sub>2</sub>, a factor that has been largely overlooked. Furthermore, it highlights a teleconnection between relatively localized terrestrial land surface cover and ocean variability that subsequently impacts climate across the North Atlantic region.



FIG. 16. Lagged correlations of the MOI with MSSA decomposed Arctic salinity anomalies at a range of depths for (top)–(bottom) 1x, 2x, 3x, and 4x. Thick dashed lines highlight 95% confidence for all time series (see section 2). Salinity anomalies lead the MOI for negative lags

#### 7. Summary and discussion

This study has used the HadCM3 coupled climate model run for millennial time scales to investigate the strength and variability of the AMOC at four quasi-equilibrium  $CO_2$  concentrations; 350, 700, 1050, and 1400 ppm (experiments 1x, 2x, 3x, and 4x, respectively). The AMOC streamfunction and an assigned AMOC index (MOI) have been decomposed and reconstructed with multivariate singular spectrum analysis (MSSA) and analyzed with EOFs in order to remove noise and isolate variability. The key findings are the following:

- AMOC strength is shown to decrease at higher concentrations of CO<sub>2</sub> with a reduction of 30.2% at 2x, 29.8% at 3x, and 34% at 4x. This is likely to be driven by a reduction in density in the Greenland–Iceland– Norwegian (GIN) Seas, an increase in stratification and so reduced convection that inhibits overturning.
- Only a small proportion of variability is explained by the first and second eigenmodes of the MSSAanalyzed AMOC streamfunctions (i.e., 11.19% at 1x).

This is increased for the SSA of the MOI (i.e., 39.53% at 1x).

- Analysis indicates that low-frequency variability of the AMOC is also weakened at higher CO<sub>2</sub>. Significant frequencies are on the order of 100–125 yr for 1x, 2x, and 3x, which increases to 160 yr at 4x, with a consecutive decrease in power with increasing CO<sub>2</sub>.
- The decline in the extent of AMOC strength and variability has consequent impacts on climate during periods of maximum and minimum AMOC.
- AMOC variability is likely to be driven primarily by salinity-driven density changes in the GIN Seas, with anomalies advected into the region predominantly from the Arctic and Caribbean regions.
- A positive salinity anomaly acts to increase density, which reduces stratification and consequently increases convection; this then acts to increase downwelling and consequently AMOC strength, increasing SSTs and SATs and decreasing SLP, which enhances Ekman pumping; and this in turn acts as a positive feedback, further enhancing convection and AMOC strength.
- In HadCM3, salinity anomalies advected from the Arctic may be key in driving variability, with the Caribbean playing a secondary role.
- Higher concentrations of  $CO_2$  are associated with a weakening in the amplitude and coherence of Arctic salinity, which in turn may be responsible for reducing the amplitude of AMOC variability. In contrast, Caribbean fluctuations remain consistent and may increase, indicating that this region may play a more predominant role at higher  $CO_2$ .
- Arctic salinity anomalies may be generated by deep northward transport from the GIN Seas as indicated by a contrasting pattern of salinity anomalies seen at depth in the Beaufort Gyre. This seesaw, depthvarying motion of anomalies may act as a positive feedback to reduce variability at higher CO<sub>2</sub> (i.e., decreased variability results in weaker northward advection, weaker anomaly generation, and consequently weaker variability). This may be a response to the overall reduced strength of the AMOC; however, further study incorporating a salinity budget and/or tracer analysis is required to test this hypothesis.
- Tropical salinity anomalies may be formed by a shift in the position of the ITCZ at 1x as identified in the study of VW04. However, at higher CO<sub>2</sub> an increase in the residence time and/or change in surface runoff may play an important role in maintaining the amplitude of anomalies. The former may be a response to a weaker overall AMOC and an increase in stratification, which increases the residence time of anomalies in the evaporative subtropical gyre. The latter may be linked to a CO<sub>2</sub>-induced vegetation dieback in the Amazon



FIG. 17. Land surface cover and runoff in the Amazon region for (top)–(bottom) 1x, 2x, 3x, and 4x. (a) Land surface cover, (b) the proportion (%) of runoff relative to precipitation for each grid box (%), and (c) composite runoff anomaly maps ( $mm day^{-1}$ ) calculated during periods of AMax averaged over periods when the PC1 of the AMOC streamfunction is one standard deviation above the mean. Only anomalies that are considered 95% confident are shown (see section 2).

region, which increases surface runoff as a proportion of precipitation. These drivers may contribute in maintaining the generation of tropical salinity anomalies that helps drive AMOC variability despite weaker Arctic fluctuations.

The modeled reduction in amplitude of variability that accompanies a weaker overall AMOC that we see in this study is in contradiction to theoretical analysis of timedependent systems, which predict an increase in variability as a system weakens and becomes more unstable (e.g., Tziperman 1997). This was first highlighted in the study of MacMartin et al. (2016) using the GFDL model and we show that this is also the case with HadCM3.

The results here highlight that AMOC variability is driven by a combination of mechanisms in HadCM3: a possible internal ocean mode in the Arctic region and a coupled atmosphere–ocean mode in the Caribbean region. Higher concentrations of  $CO_2$  act to alter these mechanisms, weakening both the Arctic and Caribbean modes and possibly driving a shift to an atmosphere–terrestrial–ocean feedback in the tropics.

This is a single-model study and some of the mechanisms presented here and how they may be influenced by  $CO_2$  are likely to be highly dependent on the model used. Inaccuracies in the simulated overflow channels and issues pertaining to resolution may have implications for the AMOC. These subsequently lead to the modeled AMOC being too shallow compared to observations, which may have implications for the time scale and the mechanisms that drive variability. Despite this, it is important and interesting to understand how variability in models may respond to climate forcing. Applying the methods outlined in this paper would elucidate how variability in other models responds to climate change. Understanding model internal variability is crucial in order to differentiate impacts that are thought to be anthropogenically forced (i.e., a CO<sub>2</sub>-induced weakening) relative to those that are a "natural" response of the modeled system. This may help us improve our ability to more accurately predict the AMOC and thus model decadal-centennial climate.

Acknowledgments. This work was funded by the U.K. National Environmental Research Council (NERC) Grant NE/L501554/1. Our thanks go to the helpful advice of two anonymous reviewers.

#### REFERENCES

- Alvarez-Garcia, F., M. Latif, and A. Biastoch, 2008: On multidecadal and quasi-decadal North Atlantic variability. *J. Climate*, **21**, 3433–3452, doi:10.1175/2007JCLI1800.1.
- Bakker, P., P. U. Clark, N. R. Golledge, A. Schmittner, and M. Weber, 2016a: Centennial-scale Holocene climate variations amplified by Antarctic Ice Sheet discharge. *Nature*, 541, 72–76, doi:10.1038/nature20582.
- —, and Coauthors, 2016b: Fate of the Atlantic meridional overturning circulation: Strong decline under continued warming and Greenland melting. *Geophys. Res. Lett.*, 43, 12 252–12 260, doi:10.1002/2016GL070457.
- Betts, R. A., P. M. Cox, M. Collins, P. P. Harris, C. Huntingford, and C. D. Jones, 2004: The role of ecosystem–atmosphere interactions in simulated Amazonian precipitation decrease and forest dieback under global climate warming. *Theor. Appl. Climatol.*, **78**, 157–175, doi:10.1007/s00704-004-0050-y.
- Boulton, C. A., P. Good, and T. M. Lenton, 2013: Early warning signals of simulated Amazon rainforest dieback. *Theor. Ecol.*, 6, 373–384, doi:10.1007/s12080-013-0191-7.
- Cattle, H., and D. Cresswell, 2000: The Arctic Ocean freshwater budget of a climate general circulation model. *Freshwater Budget of the Arctic Ocean*, E. L. Lewis et al., Eds., Springer, 127–139, doi:10.1007/978-94-011-4132-1\_6.
- Clark, P. U., N. G. Pisias, T. F. Stocker, and A. J. Weaver, 2002: The role of the thermohaline circulation in abrupt climate change. *Nature*, 415, 863–869, doi:10.1038/415863a.
- Cox, P. M., R. A. Betts, C. D. Jones, S. A. Spall, and I. J. Totterdell, 2000: Acceleration of global warming due to carbon-cycle feedbacks in a coupled climate model. *Nature*, **408**, 184–187, doi:10.1038/35041539.
- Danabasoglu, G., 2008: On multidecadal variability of the Atlantic meridional overturning circulation in the Community Climate System Model version 3. J. Climate, 21, 5524–5544, doi:10.1175/ 2008JCLI2019.1.

- —, S. G. Yeager, Y.-O. Kwon, J. J. Tribbia, A. S. Phillips, and J. W. Hurrell, 2012: Variability of the Atlantic meridional overturning circulation in CCSM4. *J. Climate*, **25**, 5153–5172, doi:10.1175/JCLI-D-11-00463.1.
- Delworth, T. L., and R. J. Greatbatch, 2000: Multidecadal thermohaline circulation variability driven by atmospheric surface flux forcing. J. Climate, 13, 1481–1495, doi:10.1175/ 1520-0442(2000)013<1481:MTCVDB>2.0.CO;2.
- —, and M. E. Mann, 2000: Observed and simulated multidecadal variability in the Northern Hemisphere. *Climate Dyn.*, 16, 661–676, doi:10.1007/s003820000075.
- —, S. Manabe, and R. J. Stouffer, 1993: Interdecadal variations of the thermohaline circulation in a coupled ocean– atmosphere model. J. Climate, 6, 1993–2011, doi:10.1175/ 1520-0442(1993)006<1993:IVOTTC>2.0.CO;2.
- —, —, and —, 1997: Multidecadal climate variability in the Greenland Sea and surrounding regions: A coupled model simulation. *Geophys. Res. Lett.*, **24**, 257–260, doi:10.1029/ 96GL03927.
- Ditlevsen, P. D., and S. J. Johnsen, 2010: Tipping points: Early warning and wishful thinking. *Geophys. Res. Lett.*, 37, L19703, doi:10.1029/2010GL044486.
- Dixon, K. W., T. L. Delworth, M. J. Spelman, and R. J. Stouffer, 1999: The influence of transient surface fluxes on North Atlantic overturning in a coupled GCM climate change experiment. *Geophys. Res. Lett.*, 26, 2749–2752, doi:10.1029/1999GL900571.
- Dong, B. W., and R. T. Sutton, 2005: Mechanism of interdecadal thermohaline circulation variability in a coupled oceanatmosphere GCM. J. Climate, 18, 1117–1135, doi:10.1175/ JCLI3328.1.
- Elsner, J. B., and A. A. Tsonis, 1996: Singular Spectrum Analysis: A New Tool in Time Series Analysis. Springer, 164 pp.
- Flato, G., and Coauthors, 2013: Evaluation of climate models. *Climate Change 2013: The Physical Science Basis*, T. F. Stocker et al., Eds., Cambridge University Press, 741–866.
- Frankcombe, L. M., and H. A. Dijkstra, 2010: Internal modes of multidecadal variability in the Arctic Ocean. J. Phys. Oceanogr., 40, 2496–2510, doi:10.1175/2010JPO4487.1.
- Gent, P. R., and J. C. McWilliams, 1990: Isopycnal mixing in ocean circulation models. J. Phys. Oceanogr., 20, 150–155, doi:10.1175/ 1520-0485(1990)020<0150:IMIOCM>2.0.CO;2.
- Ghil, M., and Coauthors, 2002: Advanced spectral methods for climatic time series. *Rev. Geophys.*, 40, 1003, doi:10.1029/ 2000RG000092.
- Golyandina, N., 2010: On the choice of parameters in singular spectrum analysis and related subspace-based methods. *Stat. Interface*, **3**, 259–279, doi:10.4310/SII.2010.v3.n3.a2.
- —, and A. Zhigljavsky, 2013: Singular Spectrum Analysis for Time Series. Springer, 120 pp.
- —, and A. Korobeynikov, 2014: Basic singular spectrum analysis and forecasting with R. *Comput. Stat. Data Anal.*, **71**, 934–954, doi:10.1016/j.csda.2013.04.009.
- —, V. Nekrutkin, and A. Zhigljavsky, 2001: Analysis of Time Series Structure: SSA and Related Techniques. Chapman and Hall/CRC, 320 pp.
- —, A. Korobeynikov, A. Shlemov, and K. Usevich, 2015: Multivariate and 2D extensions of singular spectrum analysis with the Rssa package. J. Stat. Softw., 67, 1–78, doi:10.18637/ jss.v067.i02.
- Gordon, C., C. Cooper, C. A. Senior, H. Banks, J. M. Gregory, T. C. Johns, J. F. B. Mitchell, and R. A. Wood, 2000: The simulation of SST, sea ice extents and ocean heat transports in a version of the Hadley Centre coupled model without

flux adjustments. *Climate Dyn.*, **16**, 147–168, doi:10.1007/s003820050010.

- Gregory, J. M., and Coauthors, 2005: A model intercomparison of changes in the Atlantic thermohaline circulation in response to increasing atmospheric CO<sub>2</sub> concentration. *Geophys. Res. Lett.*, **32**, L12703, doi:10.1029/2005GL023209.
- Griffies, S. M., A. Gnanadesikan, R. C. Pacanowski, V. D. Larichev, J. K. Dukowicz, and R. D. Smith, 1998: Isoneutral diffusion in a z-coordinate ocean model. J. Phys. Oceanogr., 28, 805–830, doi:10.1175/1520-0485(1998)028<0805:IDIAZC>2.0.CO;2.
- Hall, P., 1988: Theoretical comparison of bootstrap confidence intervals. Ann. Stat., 16, 927–953, doi:10.1214/aos/1176350933.
- Hawkins, E., and R. Sutton, 2007: Variability of the Atlantic thermohaline circulation described by three-dimensional empirical orthogonal functions. *Climate Dyn.*, **29**, 745–762, doi:10.1007/s00382-007-0263-8.
- —, and —, 2008: Potential predictability of rapid changes in the Atlantic meridional overturning circulation. *Geophys. Res. Lett.*, **35**, L11603, doi:10.1029/2008GL034059.
- Hu, A., and G. A. Meehl, 2005: Bering Strait throughflow and the thermohaline circulation. *Geophys. Res. Lett.*, **32**, L24610, doi:10.1029/2005GL024424.
- Huang, B. H., Z.-Z. Hu, E. K. Schneider, Z. H. Wu, Y. Xue, and B. Klinger, 2012: Influences of tropical–extratropical interaction on the multidecadal AMOC variability in the NCEP climate forecast system. *Climate Dyn.*, **39**, 531–555, doi:10.1007/ s00382-011-1258-z.
- Iwi, A. M., L. Hermanson, K. Haines, and R. T. Sutton, 2012: Mechanisms linking volcanic aerosols to the Atlantic meridional overturning circulation. J. Climate, 25, 3039–3051, doi:10.1175/2011JCLI4067.1.
- Jackson, L., and M. Vellinga, 2013: Multidecadal to centennial variability of the AMOC: HadCM3 and a perturbed physics ensemble. J. Climate, 26, 2390–2407, doi:10.1175/ JCLI-D-11-00601.1.
- Jungclaus, J. H., H. Haak, M. Latif, and U. Mikolajewicz, 2005: Arctic–North Atlantic interactions and multidecadal variability of the meridional overturning circulation. J. Climate, 18, 4013–4031, doi:10.1175/JCLI3462.1.
- Knight, J. R., 2005: A signature of persistent natural thermohaline circulation cycles in observed climate. *Geophys. Res. Lett.*, 32, L20708, doi:10.1029/2005GL024233.
- Knutti, R., and J. Sedlacek, 2013: Robustness and uncertainties in the new CMIP5 climate model projections. *Nat. Climate Change*, **3**, 369–373, doi:10.1038/nclimate1716.
- Kraus, E. B., and J. S. Turner, 1967: A one-dimensional model of seasonal thermocline II. General theory and its consequences. *Tellus*, **19**, 98, doi:10.3402/tellusa.v19i1.9753.
- Large, W. G., J. C. McWilliams, and S. C. Doney, 1994: Oceanic vertical mixing: A review and a model with a nonlocal boundary layer parameterization. *Rev. Geophys.*, **32**, 363–403, doi:10.1029/94RG01872.
- Latif, M., E. Roeckner, U. Mikolajewicz, and R. Voss, 2000: Tropical stabilization of the thermohaline circulation in a greenhouse warming simulation. J. Climate, 13, 1809–1813, doi:10.1175/1520-0442(2000)013<1809:L>2.0.CO;2.
  - —, T. Martin, and W. Park, 2013: Southern Ocean sector centennial climate variability and recent decadal trends. *J. Climate*, 26, 7767–7782, doi:10.1175/JCLI-D-12-00281.1.
- MacMartin, D. G., L. Zanna, and E. Tziperman, 2016: Suppression of Atlantic meridional overturning circulation variability at increased CO<sub>2</sub>. J. Climate, 29, 4155–4164, doi:10.1175/ JCLI-D-15-0533.1.

- Malhi, Y., and Coauthors, 2009: Exploring the likelihood and mechanism of a climate-change-induced dieback of the Amazon rainforest. *Proc. Natl. Acad. Sci. USA*, **106**, 20610– 20615, doi:10.1073/pnas.0804619106.
- Manabe, S., and R. J. Stouffer, 1997: Coupled ocean–atmosphere model response to freshwater input: Comparison to Younger Dryas event. *Paleoceanography*, **12**, 321–336, doi:10.1029/ 96PA03932.
- Marotzke, J., and P. H. Stone, 1995: Atmospheric transports, the thermohaline circulation, and flux adjustments in a simple coupled model. J. Phys. Oceanogr., 25, 1350–1364, doi:10.1175/ 1520-0485(1995)025<1350:ATTTCA>2.0.CO;2.
- Martin, T., W. Park, and M. Latif, 2013: Multi-centennial variability controlled by Southern Ocean convection in the Kiel Climate Model. *Climate Dyn.*, 40, 2005–2022, doi:10.1007/ s00382-012-1586-7.
- Menary, M. B., W. Park, K. Lohmann, M. Vellinga, M. D. Palmer, M. Latif, and J. H. Jungclaus, 2012: A multimodel comparison of centennial Atlantic meridional overturning circulation variability. *Climate Dyn.*, 38, 2377–2388, doi:10.1007/s00382-011-1172-4.
- Mignot, J., and C. Frankignoul, 2005: The variability of the Atlantic meridional overturning circulation, the North Atlantic Oscillation, and the El Ninõ–Southern Oscillation in the Bergen climate model. J. Climate, 18, 2361–2375, doi:10.1175/ JCLI3405.1.
- Mikolajewicz, U., and R. Voss, 2000: The role of the individual airsea flux components in CO<sub>2</sub>-induced changes of the ocean's circulation and climate. *Climate Dyn.*, **16**, 627–642, doi:10.1007/ s003820000066.
- Monahan, A. H., 2002: Stabilization of climate regimes by noise in a simple model of the thermohaline circulation. J. Phys. Oceanogr., 32, 2072–2085, doi:10.1175/1520-0485(2002)032<2072: SOCRBN>2.0.CO:2.
- Moron, V., R. Vautard, and M. Ghil, 1998: Trends, interdecadal and interannual oscillations in global sea-surface temperatures. *Climate Dyn.*, 14, 545–569, doi:10.1007/s003820050241.
- Msadek, R., and C. Frankignoul, 2009: Atlantic multidecadal oceanic variability and its influence on the atmosphere in a climate model. *Climate Dyn.*, 33, 45–62, doi:10.1007/s00382-008-0452-0.
- —, W. E. Johns, S. G. Yeager, G. Danabasoglu, T. L. Delworth, and A. Rosati, 2013: The Atlantic meridional heat transport at 26.5°N and its relationship with the MOC in the RAPID array and the GFDL and NCAR coupled models. J. Climate, 26, 4335–4356, doi:10.1175/JCLI-D-12-00081.1.
- Otterå, O. H., M. Bentsen, H. Drange, and L. L. Suo, 2010: External forcing as a metronome for Atlantic multidecadal variability. *Nat. Geosci.*, 3, 688–694, doi:10.1038/ngeo955.
- Pardaens, A. K., H. T. Banks, J. M. Gregory, and P. R. Rowntree, 2003: Freshwater transports in HadCM3. *Climate Dyn.*, 21, 177–195, doi:10.1007/s00382-003-0324-6.
- Park, W., and M. Latif, 2008: Multidecadal and multicentennial variability of the meridional overturning circulation. *Geophys. Res. Lett.*, **35**, L22703, doi:10.1029/2008GL035779.
- Plaut, G., and R. Vautard, 1994: Spells of low-frequency oscillations and weather regimes in the Northern Hemisphere. J. Atmos. Sci., **51**, 210–236, doi:10.1175/1520-0469(1994)051<0210: SOLFOA>2.0.CO;2.
- —, M. Ghil, and R. Vautard, 1995: Interannual and interdecadal variability in 335 years of central England temperatures. *Science*, 268, 710–713, doi:10.1126/science.268.5211.710.
- Rahmstorf, S., 1995: Bifurcations of the Atlantic thermohaline circulation in response to changes in the hydrological cycle. *Nature*, **378**, 145–149, doi:10.1038/378145a0.

- Redi, M. H., 1982: Oceanic isopycnal mixing by coordinate rotation. J. Phys. Oceanogr., 12, 1154–1158, doi:10.1175/ 1520-0485(1982)012<1154:OIMBCR>2.0.CO;2.
- Roberts, C. D., F. K. Garry, and L. C. Jackson, 2013: A multimodel study of sea surface temperature and subsurface density fingerprints of the Atlantic meridional overturning circulation. *J. Climate*, 26, 9155–9174, doi:10.1175/JCLI-D-12-00762.1.
- Roberts, M. J., R. Marsh, A. L. New, and R. A. Wood, 1996: An intercomparison of a Bryan–Cox-type ocean model and an isopycnic ocean model. Part I: The subpolar gyre and highlatitude processes. J. Phys. Oceanogr., 26, 1495–1527, doi:10.1175/1520-0485(1996)026<1495:AIOABT>2.0.CO;2.
- Schaller, N., I. Mahlstein, J. Cermak, and R. Knutti, 2011: Analyzing precipitation projections: A comparison of different approaches to climate model evaluation. J. Geophys. Res., 116, D10118, doi:10.1029/2010JD014963.
- Schmittner, A., C. Appenzeller, and T. F. Stocker, 2000: Enhanced Atlantic freshwater export during El Niño. *Geophys. Res. Lett.*, 27, 1163–1166, doi:10.1029/1999GL011048.
- Smeed, D., G. McCarthy, D. Rayner, B. I. Moat, W. E. Johns, M. Baringer, and C. S. Meinen, 2015: Atlantic meridional overturning circulation observed by the RAPID-MOCHA-WBTS (RAPID-Meridional Overturning Circulation and Heatflux Array-Western Boundary Time Series) array at 26°N from 2004 to 2014. British Oceanographic Data Centre, Natural Environment Research Council, doi:10.5285/ 35784047-9b82-2160-e053-6c86abc0c91b.
- Stenchikov, G., T. L. Delworth, V. Ramaswamy, R. J. Stouffer, A. Wittenberg, and F. R. Zeng, 2009: Volcanic signals in oceans. J. Geophys. Res., 114, D16104, doi:10.1029/ 2008JD011673.
- Stommel, H., 1961: Thermohaline convection with two stable regimes of flow. *Tellus*, **13**, 224–230, doi:10.3402/tellusa.v13i2.9491.
- Swingedouw, D., P. Braconnot, P. Delecluse, E. Guilyardi, and O. Marti, 2007: Quantifying the AMOC feedbacks during a 2×CO<sub>2</sub> stabilization experiment with land-ice melting. *Climate Dyn.*, **29**, 521–534, doi:10.1007/s00382-007-0250-0.
- —, J. Mignot, S. Labetoulle, E. Guilyardi, and G. Madec, 2014: Initialisation and predictability of the AMOC over the last 50 years in a climate model. *Climate Dyn.*, **42**, 555–556, doi:10.1007/s00382-013-2005-4.
- Thorpe, R. B., 2005: The impact of changes in atmospheric and land surface physics on the thermohaline circulation response to anthropogenic forcing in HadCM2 and HadCM3. *Climate Dyn.*, 24, 449–456, doi:10.1007/s00382-004-0493-y.
- —, J. M. Gregory, T. C. Johns, R. A. Wood, and J. F. B. Mitchell, 2001: Mechanisms determining the Atlantic thermohaline circulation response to greenhouse gas forcing in a non-flux-adjusted

coupled climate model. J. Climate, 14, 3102–3116, doi:10.1175/ 1520-0442(2001)014<3102:MDTATC>2.0.CO:2.

- Timmermann, A., M. Latif, R. Voss, and A. Grotzner, 1998: Northern Hemispheric interdecadal variability: A coupled air–sea mode. J. Climate, 11, 1906–1931, doi:10.1175/1520-0442-11.8.1906.
- Toggweiler, J. R., and B. Samuels, 1993: Is the magnitude of the deep outflow from the Atlantic Ocean actually governed by Southern Hemisphere winds? *The Global Carbon Cycle*, M. Heimann, Ed., Springer, 303–331, doi:10.1007/978-3-642-84608-3\_13.
- Tziperman, E., 1997: Inherently unstable climate behaviour due to weak thermohaline ocean circulation. *Nature*, **386**, 592–595, doi:10.1038/386592a0.
- —, J. R. Toggweiler, Y. Feliks, and K. Bryan, 1994: Instability of the thermohaline circulation with respect to mixed boundary conditions: Is it really a problem for realistic models? *J. Phys. Oceanogr.*, 24, 217–232, doi:10.1175/1520-0485(1994)024<0217: IOTTCW>2.0.CO;2.
- Valdes, P. J., and Coauthors, 2017: The BRIDGE HadCM3 family of climate models: HadCM3@Bristol v1.0. *Geosci. Model Dev.*, doi:10.5194/gmd-2017-16, in press.
- Vautard, R., P. Yiou, and M. Ghil, 1992: Singular-spectrum analysis: A toolkit for short, noisy chaotic signals. *Physica D*, 58, 95–126, doi:10.1016/0167-2789(92)90103-T.
- Vellinga, M., and P. L. Wu, 2004: Low-latitude freshwater influence on centennial variability of the Atlantic thermohaline circulation. J. Climate, 17, 4498–4511, doi:10.1175/3219.1.
- von Storch, J. S., P. Müller, R. J. Stouffer, R. Voss, and S. F. B. Tett, 2000: Variability of deep-ocean mass transport: Spectral shapes and spatial scales. J. Climate, 13, 1916–1935, doi:10.1175/ 1520-0442(2000)013<1916:VODOMT>2.0.CO;2.
- Wiesenfeld, K., and B. McNamara, 1986: Small-signal amplification in bifurcating dynamical systems. *Phys. Rev.*, 33A, 629– 642, doi:10.1103/PhysRevA.33.629.
- Wilks, D. S., 1997: Resampling hypothesis tests for autocorrelated fields. J. Climate, 10, 65–82, doi:10.1175/1520-0442(1997)010<0065: RHTFAF>2.0.CO;2.
- Wunsch, C., and P. Heimbach, 2006: Estimated decadal changes in the North Atlantic meridional overturning circulation and heat flux 1993–2004. J. Phys. Oceanogr., 36, 2012–2024, doi:10.1175/JPO2957.1.
- Zanchettin, D., C. Timmreck, H. F. Graf, A. Rubino, S. Lorenz, K. Lohmann, K. Kruger, and J. Jungclaus, 2012: Bi-decadal variability excited in the coupled ocean–atmosphere system by strong tropical volcanic eruptions. *Climate Dyn.*, **39**, 419–444, doi:10.1007/s00382-011-1167-1.
- Zhu, X., and J. Jungclaus, 2008: Interdecadal variability of the meridional overturning circulation as an ocean internal mode. *Climate Dyn.*, **31**, 731–741, doi:10.1007/s00382-008-0383-9.